Iowa State University 1 CprE489, Lecture 8

CprE 489 Lecture 8

Fall Semester 1997

Iowa State University 2 CprE489, Lecture 8

Packet Switching: Time Analysis

packet time propogation time processing time packet time = p/b prop time ~ distance proc time ~ queuing

Time for first bit to arrive = (h-1)(p/b + proc time) + h(prop time) h = number of hops p = packet length b = bit rate

Iowa State University 3 CprE489, Lecture 8

Pipelining of bits in a message

Total time = time for first bit to arrive + sum of packet times = (h-1)(p/b + proc time) + h(prop time) + kp/b

kp = message size

Iowa State University 4 CprE489, Lecture 8

Pipelining of packets in a message

Total time = time for first (k-1) packets + time for the last packet = (k-1)*(d) + h*(pc+p/b+pp) Assume d > pc + p/b

kp = message size pc pp

d p/b

or d = pc + p/b + a, then = (h-1)*(pc+p/b) + h*pp + pc + p/b + (k-1)*(pc+p/b+a) =(h-1)*(pc+p/b) + h*pp + kp/b + (k-1)*(pc+a) + pc In Slide 3, a= -pc, and Initial pc is ignored